

Manuscript No: 59800

Dear Editor of World Journal of Clinical Cases.

I would like to re-submit the revised manuscript titled "**Unilateral pulmonary hemorrhage caused by negative pressure pulmonary edema. -A case report and review of literature-**" Hyung-Joon Park, Seong-Ho Park, Un-Tak Woo, Sang-Yun Cho, Woo-Jae Jeon and myself to World Journal of Clinical Cases.

We revised a manuscript according to the reviewer's recommendation. Even though radiologic changes were definite during the course of admission in the intensive care unit, pathologic changes on the lung were not clear and we described it and a probable reason of the unilateral alveolar hemorrhage in the discussion section.

We checked if the author's names and institutions meet the requirements of the journal. We added the "FINAL DIAGNOSIS", "TREATMENT", and "OUTCOME AND FOLLOW-UP" sections to the case presentation section according to the Guidelines and Requirements for Manuscript Revision. We prepared and re-arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor. Abstract, Main Text, and Acknowledgements was further revised according to the guidelines for revising the content. References were revised according to the [Format for references guidelines](#) using the reference auto-analyser.

We assessed the importance anesthesiologist's role in postoperative period. Anesthesiologists should take care not to induce airway irritation during emergence of asthma, because diffuse alveolar hemorrhage accompanied by NPPE was caused in this case by airway irritation in an older patient with asthma without known risk factors.

I hope you will consider this report as suitable for accept in the World Journal of Clinical Cases. We corrected our article which reviewers recommended to us and updated our manuscript

I'm looking forward to your reply. Thank you.

Sincerely yours

Woo-Jong Shin MD, PhD, Professor

Address: #153 Gyeongchoon-ro, Department of Anesthesia and Pain Medicine,  
Hanyang University Guri Hospital, Guri-si, Gyeonggi-do, 11923, Republic of Korea

Tel.: +82-31-560-2400; Fax: +82-31-563-1731

E-mail: swj0208@hanyang.ac.kr